

JGR Solid Earth

RESEARCH ARTICLE

10.1029/2021JB022377

Key Points:

- Geodynamic and seismological modeling methods are combined to reproduce the recent evolution of the Central Mediterranean
- The influence on trench shape and slab tears of structural heterogeneities within the Adria plate is addressed
- Geodynamic models are validated against seismic anisotropy observations

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

M. Faccenda, manuele.faccenda@unipd.it

Citation:

Lo Bue, R., Faccenda, M., & Yang, J. (2021). The role of Adria plate lithospheric structures on the recent dynamics of the Central Mediterranean region. *Journal of Geophysical Research: Solid Earth*, *126*, e2021JB022377. https://doi.org/10.1029/2021JB022377

Received 9 MAY 2021 Accepted 20 SEP 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

The Role of Adria Plate Lithospheric Structures on the Recent Dynamics of the Central Mediterranean Region

Rosalia Lo Bue¹, Manuele Faccenda¹, and Jianfeng Yang¹

¹Dipartimento di Geoscienze, Università di Padova, Padova, Italy

Abstract The Tertiary tectonic evolution of the Central Mediterranean has been relatively well constrained by abundant geological data. Yet, several uncertainties persist about the mechanisms that led to the present-day surface morphology and deep slab geometry. Here, we combine geodynamic and seismological numerical modeling techniques to reproduce the recent large-scale evolution of the Central Mediterranean and the associated strain-induced upper mantle fabrics and seismic anisotropy. 3D thermo-mechanical subduction models were designed and calibrated according to paleogeographic-tectonic reconstructions and seismological observations available in the literature. It is found that, although the opening of back-arc extensional basins in response to the retreat of the Ionian slab is a common feature in all models, structural heterogeneities within the Adria plate and/or the geometry of its Tyrrhenian passive margin profoundly impact on the segmentation of the subducting slab and the amount of Ionian trench retreat. More, in general, this study highlights the importance of coupling geodynamic and seismological modeling to better constrain the tectonic evolution of complex convergent margins such as the Central Mediterranean.

Plain Language Summary The Central Mediterranean represents one of the most complex subduction zones on Earth. Its recent tectonic evolution is linked to the motion of three main plates (Africa, Adria, and Europe) and to the opening of the Liguro-Provençal and Thyrrenian basins. However, the mechanisms that led to the present-day surface and deep morphology of the region remain unclear. Here, we present a new approach, which can reduce these uncertainties by combining geodynamic and seismological numerical modeling techniques. Our study confirms that the main geological and geophysical observables in the Central Mediterranean are likely linked to the recent dynamics of the Tyrrhenian slab and the Adria plate. In addition, we show that structural heterogeneities within the Adria plate play an important role in the formation of a wide slab window below the Central Apennines, in the retreat of the Northern Apennines and the Ionian trenches till the present-day position, and in the curvature of the Apennines chain.

1. Introduction

In the geodynamic context of the slow convergence between the African and Eurasian plates, the Central Mediterranean region has been involved in a complex subduction process that over the last 30 Myr was characterized by the rapid retreat of the Ionian trench, the opening of back-arc extensional basins (i.e., Liguro-Provençal, Algerian, and Tyrrhenian basins) and episodes of slab lateral tearing, segmentation, and break-off (Carminati et al., 2012; Faccenna et al., 2004, 2007, 2014; Jolivet et al., 2006, 2009; Mauffret et al., 2004; Rosenbaum et al., 2002b; van Hinsbergen et al., 2014, 2020). Although the shallow tectonic evolution has been relatively well constrained by a wealth of geological data, several uncertainties persist about the recent mantle dynamics of this region and the interaction between surface tectono-magmatic processes and deep mantle convection patterns. Analog and/or numerical modeling of 3D mantle flow evolution specifically designed for the Mediterranean could provide important clarifications about these processes. However, given the high complexity of the Mediterranean tectonic setting, few research groups have attempted to reproduce its evolution.

On the one hand, analog models of a single subducting and rolling back slab have been able to address and reproduce some of the key aspects of the Central Mediterranean history such as (a) the initiation of the subduction process, (b) the subduction dynamics and their relationships with plate kinematics, (c) the episodicity of back-arc extension, and finally (d) the role played by slab-induced mantle circulation (Funiciello

& Faccenna, 2010; Guillaume et al., 2010, 2013). On the other hand, 3D numerical models have highlighted the importance of considering appropriate plate geometries and lateral buoyancy variations to reproduce the recent dynamics of the Central-Western Mediterranean and Alpine regions (Kaus et al., 2020; Lo Bue & Faccenda, 2019; Magni et al., 2014).

Among the aspects of the Central Mediterranean geodynamic history that have not yet been fully clarified, there are (a) the development and evolution of the episodes of slab tearing and segmentation along northern Africa and the western Adria plate margins, and (b) the relations between mantle dynamics and plate surface and deep kinematics. The latter can be estimated by measuring seismic anisotropy generated by strain-induced lattice/crystal preferred orientation (LPO/CPO) of intrinsically anisotropic minerals, which is thought to be the primary source of seismic anisotropy (Park & Levin, 2002). As there is a direct link between upper mantle deformation and macroscopic seismic anisotropy, a quantitative analysis of the seismic anisotropy may help to reconstruct the recent upper mantle flow patterns.

Laboratory experiments indicate that the strain-induced LPO patterns vary as a function of the deformation history, temperature, deviatoric stress, and water content conditions (Karato et al., 2008). It follows that the extrapolation of the mantle flow from seismic anisotropy is neither simple nor always warranted, especially at subduction zones where complex and non-steady-state 3D flow patterns may establish. A promising approach that helps to reduce the number of plausible models that can explain a given anisotropy data set is to compare seismic measurements with predictions of numerical and experimental flow models (Long et al., 2007). Traditional numerical modeling studies of mantle fabrics and the associated anisotropy assume time-independent, steady-state flow dynamics (Becker, 2006). This method is effective in global intra-oceanic contexts (e.g., Pacific and Atlantic oceans) where plate motions and shallow mantle flow remained stable over the past 43 Ma (Faccenna et al., 2013). Recently, Faccenda and Capitanio (2012, 2013) have extended this methodology to account for the non-steady-state evolution typical of many subduction zones, yielding mantle fabrics that are physically consistent with the deformation history. Hu et al. (2017), Zhou et al. (2018), and Confal et al. (2018) applied a similar modeling approach to real tectonic settings, such as the South America, the North America, and the eastern Mediterranean subduction systems, respectively.

In this study, we first briefly review the Central Mediterranean tectonic evolution and the seismological data available for the region. Subsequently, we show the results of applying the modeling methodology of Faccenda and Capitanio (2012, 2013) to the Central Mediterranean convergent margin. The modeling results are then evaluated and discussed by comparing seismological synthetics (SKS splitting, Rayleigh and P-wave anisotropy, and isotropic anomalies) and the predicted tectonic evolution with geophysical and geological observations. Throughout the study, we use "Ma" to indicate geological time before present, and "Myr" when referring to the elapsed time after the start of the model.

1.1. Tectonic Evolution of the Region

The tectonic evolution of the entire Mediterranean region in the Mid-Late Cenozoic is connected with the relative motion of three main plates (Africa, Adria, and Europe) and in its central portion is dominated by the collision between the Adria and Eurasian plates along the Alps. Two oceanic trenches flanked the Alpine collision: the Liguro-Provençal/Tyrrhenian trench on the western side (object of this study) and the Hellenic trench on the eastern side. The presence of an unknown number of smaller continental blocks and oceanic basins in the Tethyan realm produced intermittent phases of subduction and collision that led to the currently observed complex geological setting (Carminati et al., 2012; Faccenna et al., 2014).

Before ~30–35 Ma, the underthrusting of the European continental lithosphere below Apulia-Adria induced the formation of the Alps. After ~30–35 Ma, when the subduction rate of the Ionian ocean overcame the Africa-Eurasia convergence rate, the tectonic regime along the oceanic trenches switched from compressional to extensional. The evolution of this later tectonic stage (Figure 1a) can be summarized in several episodes of slab roll-back and opening of back-arc basins (Carminati et al., 2012; Dewey et al., 1989; Gueguen et al., 1998; Faccenna et al., 2004, 2007, 2014; Jolivet et al., 2009; Malinverno & Ryan, 1986; Rosenbaum et al., 2002a; Wortel & Spakman, 2000).

Several widely accepted tectonic reconstructions indicate that at ~35–30 Ma an incipient Ionian slab was already present in the upper mantle below the Corsica-Sardinia continental block, which may have triggered





Figure 1. (a) Evolution of the Central Mediterranean region in the last 30 Myr (adapted from Lucente & Speranza, 2001, Lucente et al., 2006, Faccenna et al., 2014, and van Hinsbergen et al., 2014). Red lines indicate active subduction at different age stages, while the black thick line is the present-day frontal thrust. The brown region illustrates the geometry of the Africa and Adria continental plates, excluding the thin margins, at 30 Ma (adapted from Lucente & Speranza, 2001, Lucente et al., 2006, and van Hinsbergen et al., 2014). (b) SKS-splitting data in the Central-Western Mediterranean (Becker et al., 2012) and references therein updated as of December 6, 2020 (http://www-udc.ig.utexas.edu/external/becker/sksdata). The bars show the orientation of the SKS fast component and their length and color-coding is proportional to the delay time.

the subsequent "spontaneous" evolution of the subduction system (Faccenna et al., 2001, 2014). In the first stage of back-arc extension, from ~32–30 Ma to ~16–15 Ma, Corsica and Sardinia were separated from the European mainland by a ~40° clockwise rotation, leading to the opening of the Liguro-Provençal basin (Faccenna et al., 1997, 2014; Gattacceca & Speranza, 2002; Rollet et al., 2002; Rosenbaum et al., 2002b). The collision between the Corsica-Sardinia microplate with the western margin of Adria caused the formation of the Apennines (Patacca et al., 1990).

The second major stage of trench retreat, from ~15 Ma to the present-day, led to the opening of the Tyrrhenian Sea basin. The first period of extension was characterized by continental lithosphere thinning in the northern Tyrrhenian Sea, and by trench retreat in the western part of the southern Tyrrhenian Sea. In the latest Messinian (6–5 Ma), the ultrafast roll-back rate of the narrow Ionian slab toward the southeast induced the opening of the southern Tyrrhenian Sea basin and the formation of new oceanic lithosphere. This happened in two distinct episodes with the formation of the Vavilov Basin (4–3 Ma) and, further to the east, of the Marsili Basin (2–1 Ma) surrounded by the currently active Aeolian island arc (Faccenna et al., 2007).

1.2. Isotropic Seismic Tomographies

Since the late 1990s, seismic tomographic methods have revealed several positive velocities anomalies interpreted as subducted lithospheric portions lying below the Alboran, Kabylides, and Calabria-Apennine areas (Bezada et al., 2013; Gutscher et al., 2002; Piromallo & Morelli, 2003; Spakman, 1991; Spakman et al., 1993; Spakman & Wortel, 2004; Van der Meer et al., 2018; Wortel & Spakman, 2000; Wortel et al., 2009).

A dominant feature in P-wave tomographic models is a broad high velocity anomaly associated with the North Apenninic-Calabrian slab (Amato et al., 1993; Lucente et al., 1999; Piromallo & Morelli, 2003; Selvag-gi & Chiarabba, 1995; Spakman et al., 1993; Spakman & Wortel, 2004; Van der Meer et al., 2018).

To the south, the narrow Calabria slab appears to broaden at the base of the upper mantle and to stagnate in the mantle transition zone of the Central Mediterranean (Neri et al., 2009; Spakman & Wortel, 2004). A similar result was obtained by Giacomuzzi et al. (2012) using a tomographic inversion based on analysis of relative residuals of S-wave arrivals from teleseismic events, and by Scarfi et al. (2018), who detected a



narrowing of the subduction system caused by horizontal tearing affecting both the sides of the slab. Neri et al. (2012) analyzed different geophysical data sets (gravity anomalies, seismic tomographic structure, and seismicity within the crust and uppermost mantle) and proposed that the slab is still continuous beneath the central part of the Calabrian arc, whereas detachment has already occurred beneath the arc edges (beneath northern Calabria and northeastern Sicily). Presti et al. (2019) suggested that clear necking of the foundering slab has already occurred (at a depth of 150 km), slab detachment has not happened yet but appears to be geologically close, implying the imminent end of the subduction process in the Central Mediterranean.

Further north along the Apenninic chain, several seismic tomography studies associated high-velocity anomalies to the subducted Ionian ocean and western margin of the Adria plate (Benoit et al., 2011; Giacomuzzi et al., 2012; Kästle et al., 2018; Koulakov et al., 2009; Lucente et al., 1999; Piromallo & Morelli, 2003; Rosenbaum et al., 2008; Spakman & Wortel, 2004; Van der Meer et al., 2018; Zhao et al., 2016). The same studies indicate that the high-velocity anomalies found below the Calabrian arc and the Northern Apennines appear to be laterally continuous below 250 km depth. At shallower depths, instead, a slow velocity anomaly is found below central Italy, which is associated with the presence of a slab window.

The vertical continuity and lateral extent of the slab beneath the northern Apennines are debated. Based on early tomographic models showing no apparent connection between the Adriatic plate and the subducted lithosphere, Spakman (1990) and Spakman et al. (1993) interpreted the northern Apennines deep anomaly as a detached slab under the entire Apennines resulting from the lateral migration of a slab tear from north to south. Conversely, the subsequent regional mantle models of Piromallo and Morelli (1997, 2003) exhibit local evidence for a continuous slab beneath Tuscany, at the same location where Spakman et al. (1993) found a small slab gap only about 50 km wide. Other teleseismic tomography studies of the Italian peninsula do not reveal any significant gap in the subducted lithosphere below the northern Apennines, and suggest the presence of a slab window below the central-southern Apennines and in between the two high-velocity bodies in the northern Apennines and Calabrian regions (Amato et al., 1993; Lucente et al., 1999). Following these results, Lucente and Speranza (2001), proposed an evolution of the Apennines arc-trench system strongly influenced by the irregular shape of the Adria continental margin colliding with the trench. According to this study, the cessation of subduction and the detachment of the slab beneath central Italy was possibly caused by the presence of a thick continental lithosphere promontory of the Adria plate. However, more recent studies suggest that the northern Apennines slab reaches a depth of only ~150-300 km (Giacomuzzi et al., 2012; Spakman & Wortel, 2004).

El-Sharkawy et al. (2020) used Rayleigh wave tomography to image the entire Mediterranean upper mantle down to about 300 km. Although Rayleigh wave tomographies are characterized by a low lateral resolution and are affected by vertical smearing (strictly speaking, also teleseismic body wave tomographies of the upper mantle bear the same problem), they provide a useful source of information about the structure of shallower mantle layers that supplement those derived from body wave models. The El-Sharkawy et al. (2020)'s model highlights that the Calabrian slab is vertically continuous in the area beneath Calabria and easternmost Sicily but detached beneath the southern Apennines and the northern Sicily. In particular, a low-velocity anomaly extending from 70 km down to about 250 km depth is detected beneath the central Apennines, which supports the presence of the Central Apennines Slab Gap. Below 250 km depth, a high-velocity anomaly is detected below the central Apennines, which can be related to the top edge of the north-westward dipping Calabrian Slab. Further north, the high-velocity anomaly appears to extend down to only 250 km depth beneath the Northern Apennines.

1.3. Seismic Anisotropy

Seismic anisotropy is widespread in the Mediterranean and it shows a complex pattern that likely has some relations with the recent (20–30 Ma) tectonic evolution of this region (Buontempo et al., 2008; Lucente et al., 2006; Schmid et al., 2004). Measurements of shear wave splitting of core-refracted (mainly SKS and SKKS, herein referred for simplicity to just SKS) waves in the Central Mediterranean is characterized by average delay times of around 1–2 s, suggesting the presence of well-developed upper mantle fabrics and strong seismic anisotropy. The SKS fast azimuths have a preferential trend of NNW-SSE (i.e., trench parallel) along the Apenninic belt, which is turning to a more NS orientation over the Apennines foreland (Bac-



cheschi et al., 2007; Civello & Margheriti, 2004; Lucente et al., 2006; Lucente & Margheriti, 2008; Margheriti et al., 2003; Petrescu et al., 2020; Plomerová et al., 2006; Salimbeni et al., 2013) (Figure 1b). The trench-parallel anisotropy appears to be in contrast with the trench-perpendicular P-wave fast directions found by Hua et al. (2017), in correspondence of the isotropic high-velocity anomaly imaged beneath the Northern Apennines as described in Section 1.2. Moving westward over the Tyrrhenian basin, SKS delay times increase and the fast azimuths, parallel to the stretching direction of the back-arc basin, have been related to back-arc extension (Faccenna et al., 2014; Jolivet et al., 2009; Lucente et al., 2006). More in general, EW-oriented fast directions running parallel to the opening trajectory of the Liguro-Provençal and Tyrrhenian basins are observed in the Central Mediterranean by both SKS and Rayleigh waves (Zhu & Tromp, 2013). Beneath the central Apennines, instead, smaller delay times have been interpreted as being due to a larger vertical component of shear and related steeply oriented fast axes (Lucente & Margheriti, 2008). The presence of trench-perpendicular SKS fast azimuths has been related to either subduction windows or to toroidal mantle flow patterns at the edge of slabs. An example of the latter feature can be found in the Sicily channel where SKS fast axes turn to NS and then EW in the Tyrrhenian Sea (Baccheschi et al., 2007; Civello & Margheriti, 2004).

2. Methods

In this section, we describe the numerical methods used to generate subduction-induced mantle flow patterns and to calculate strain-induced mantle fabrics and synthetic seismic anisotropy.

2.1. Mechanical Numerical Modeling

3D petrological-thermo-mechanical models of subduction have been carried out with I3MG that is based on the finite difference method (FDM) combined with a marker-in-cell (MIC) technique (Gerya, 2019). The physical properties are defined on and advected by Lagrangian markers. A staggered Eulerian grid is defined to solve the equations of conservation of mass (Equation 1), momentum (Equation 2), and energy (Equation 3):

$$\nabla \cdot \vec{u} = 0 \tag{1}$$

$$-\nabla P + \nabla \cdot \tau = -\rho \vec{g} \tag{2}$$

$$\rho C_p \frac{DT}{Dt} = -\nabla \cdot \vec{q} + H \tag{3}$$

where D/Dt is the material time derivative and $\vec{u}, P, \tau, \rho, C_p, \vec{g}, T, \vec{q}$ are the velocity, pressure, deviatoric stress, density, heat capacity, gravitational acceleration ($g_x = g_z = 0$, $g_y = 9.81 \text{ ms}^{-2}$), temperature, and heat flux, respectively. The heat source term H accounts for radiogenic, adiabatic, and shear heating. The stable mineralogy and physical properties for the mantle used in our models are those computed with PERPLE_X (Connolly, 2005) and tested by Mishin et al. (2008) for a pyrolytic mantle composition. One of the main effect of taking thermodynamic databases into account is the incorporation of solid-state phase transitions, with uncertainties on mineral thermodynamic properties that increase with depth. These transitions are associated with significant changes in mantle density and seismic wave speeds (Turcotte & Schubert, 2014), and affect the dynamics of mantle convection due to (a) density changes and (b) latent heating (Christensen & Yuen, 1985; Richter, 1973; Schubert et al., 1975; Tackley, 1993; Zhong & Gurnis, 1994). In our numerical models, the only major phase transition is the olivine-spinel occurring at about 410 km depth (the spinel-perovskite reaction at 670 km depth does not affect at all the model evolution that confined in the upper mantle and transition zone). Although the slab has a more depleted composition, it has been demonstrated that the density anomalies of a harzburgitic mantle are similar to those of a pyrolitic composition (Faccenda & Dal Zilio, 2017). This confirms what was found by the study of Nakagawa et al. (2009), which showed that even at global scale when using phase assemblages with different major oxides abundance and computed with the same software (PERPLE_X) the mantle convection patterns do not change substantially.

The numerical domain is a cartesian box defined by the (x - y - z) coordinates (2600 × 700 × 1800 km) discretized with (261 × 101 × 181 nodes), with *y* being the vertical direction. Velocity boundary conditions



Table 1

DI		D. 1. TT.	1	
Physical Pro	pernes of	ROCKS USE	a in 1n	is Siuay

Property	Symbol	Value	Unit				
Diffusion creep (Karato & Wu, 1993)							
Pre-exponential factor	А	$8.7\cdot 10^{15}$	s^{-1}				
Activation energy	Е	300	kJmol ⁻¹				
Activation volume	V	4.5	$\mathrm{cm}^3\mathrm{mol}^{-1}$				
Stress exponent	n	1	-				
Grain-size exponent	m	2.5	-				
Burger vector	b	0.5	nm				
Grain size	d	1	mm				
Dislocation creep (Karato & Wu, 1993)							
Pre-exponential factor	А	$3.5\cdot 10^{22}$	s^{-1}				
Activation energy	Е	540	kJmol ⁻¹				
Activation volume	V	17	$\rm cm^3 mol^{-1}$				
Stress exponent	n	3.5	-				
Grain-size exponent	m	0	-				
Peierls creep (Katayama & Karato, 2008)							
Pre-exponential factor	А	10 ^{7.8}	Pa ² s				
Activation energy	Е	532	kJmol ⁻¹				
Activation volume	V	12	cm ³ mol ⁻¹				
Peierls stress	$\sigma_{\it Peierls}$	9.1	GPa				
Exponent	p, q	1,2	-, -				

Note. $R = 8.313 \text{ Jmol}^{-1}\text{K}^{-1}$ is the gas constant, while $\mu = 80$ GPa is the shear modulus.

are free slip everywhere. We focus on modeling self-consistent subduction driven by internal buoyancy forces (no kinematic conditions are prescribed). A constant temperature of 273 K is applied at the top boundary, while a constant incoming heat flux of 2 mW/m^2 is imposed at the bottom boundary. The side boundaries are insulating.

The mantle mechanical behavior is modeled using a visco-plastic rheology based on deformation invariants (Ranalli, 1995). The effective viscosity is given by the harmonic average of the combined dislocation, diffusion, and Peierls creep mechanisms (parameters and physical meaning are defined in Table 1):

$$\eta_{ductile} = \left(\frac{1}{\eta_{disl}} + \frac{1}{\eta_{diff}} + \frac{1}{\eta_{peierls}}\right)^{-1}$$
(4)

where the dislocation and diffusion creep are given by (Karato & Wu, 1993):

$$\dot{\varepsilon} = A \left(\frac{\sigma}{\mu}\right)^n \left(\frac{b}{d}\right)^m exp\left(-\frac{E+PV}{RT}\right)$$
(5)

$$\eta = \frac{\sigma}{2\dot{\varepsilon}} \tag{6}$$

At elevated stresses (0.1 GPa) typical of low-T conditions creep is accommodated via the Peierls mechanism as (Katayama & Karato, 2008):

$$\eta_{peierls} = 0.5A\sigma_{II}^{'-1}exp\left\{\frac{E+PV}{RT}\left[1-\left(\frac{\sigma_{II}^{'}}{\sigma_{peierls}}\right)^{p}\right]^{q}\right\}.$$
 (7)

A pseudo-plastic viscosity is computed as:

$$\eta_{pl} = \frac{\tau_y}{2\dot{\varepsilon}_{II}} \tag{8}$$

Finally, the effective viscosity is given by:

$$\eta_{eff} = min(\eta_{ductile}, \eta_{pl}) \tag{9}$$

The plastic strength τ_v is determined with a plastic Drucker-Prager criterion (Ranalli, 1995):

$$\tau_{v} = C_{DP} + \mu P \tag{10}$$

where $C_{DP} = C \cos \phi = 1MPa$ is the cohesion, $\mu = \sin \phi$ is the friction coefficient, and ϕ is the friction angle. To model strain-induced brittle weakening, the initial friction coefficient is linearly decreased upon reaching a final lower value when the accumulated plastic strain is 0.1. The lower and upper cutoff of the viscosity are set to 10^{18} and 10^{24} Pa s.

The geometry of the Central Mediterranean was drawn according to the paleogeographic and tectonic reconstructions at \sim 30 Ma proposed by Lucente and Speranza (2001), Lucente et al. (2006), Faccenna et al. (2014), and van Hinsbergen et al. (2014). The 3D initial temperature and compositional (crust vs. mantle) fields were created with the MATLAB toolbox geomIO (Bauville & Baumann, 2019) and then imported on I3MG to define the initial setup of the petrological-thermo-mechanical model.

The initial numerical model (Figure 2) consists of a subducting oceanic plate that represents the Ionian Ocean, two lateral continental blocks corresponding to the Adria and Africa plates, and an overriding plate that corresponds to the Iberia and European plates. To initiate subduction self-consistently, (a) a wide slab is located from Gibraltar to Corsica dipping 40° and extending down to 300 km in the upper mantle, (b) a rheologically weak zone has been inserted on the slab top surface to lubricate the initial contact between the overriding and the subducting plates (constant viscosity of $10^{18} Pa s$ and constant density of 3200 kg/m^3), and (c) the overriding plate in between the trench and the European plate is composed of a young lithospheric portion that offers little resistance to slab roll-back and trench





Figure 2. Initial model setup for the reference Model CM. It consists of a subducting oceanic plate (Ionian Ocean; light blue), two lateral continental blocks (Adria and Africa; orange), an overriding plate (Iberia and Europe: orange; thin upper plate: light gray) drawn according to paleogeographic reconstructions at 30 Ma proposed by Faccenna et al. (2014) and van Hinsbergen et al. (2014). The Adria plate is characterized by the presence of a stiffer continental promontory (dark orange) and a thin continental margin (red) as proposed by Lucente and Speranza (2001) and Lucente et al. (2006). A wide slab is located from Gibraltar to Corsica extending down to 300 km. The solid black line indicates the coastlines at 30 Ma (van Hinsbergen et al., 2014), while the dashed black line the present-day coastlines of peninsular Italy.

retreat. The 300 km long slab is needed to trigger slab roll-back self-consistently, and it might be representative of a more recent stage of the Central Mediterranean history rather than the 30 Ma assumed here. In some models, the Adria plate structure, located in the current Apenninic area, is characterized by the presence of a stiffer continental promontory in its central portion and of a thin continental lithosphere in the Umbria-Marche region (Calcagnile & Panza, 1980; Geiss, 1987; Lucente & Speranza, 2001; Lucente et al., 2006; Maino et al., 2013; Miller & Piana Agostinetti, 2012; Panza et al., 2003). The African plate consists of a slightly thinner margin on its eastern side toward the Ionian ocean (Arab et al., 2020). The initial lithosphere temperature distribution has been determined according to the half-space cooling equation (Turcotte & Schubert, 2014), while the underlying asthenosphere is characterized by a constant adiabatic temperature gradient of 0.5 Kkm⁻¹. The continental plates (Africa, Africa eastern margin, Iberia, Adria, and Adria promontory) are set to be 150 Myr, the western-northern portion of Adria is 90 Myr old to replicate a thin continental lithosphere, the young portion of the upper plate is 1 Myr to facilitate the trench retreat (we justify the young age by assuming a well-developed continental rifting system North of the Balearics and Corsica-Sardinia block). The subducting oceanic plate is 80 Myr old (which is representative of the thermal structure of the Piemonte-Ligurian oceanic lithosphere 30 Ma ago; van Hinsbergen et al., 2014), while the slab in the mantle has an age of 70 Myr to model partial heating up by surrounding mantle. The density is calculated using the thermodynamic databases except in some models where for the crust of the continental promontory we use a constant value of 2700 kg/m³. Otherwise, the continental crust density is computed as being that of the mantle minus 400 kg/m³, except for the Adria thin margin where we subtract 200 kg/m^3 to model a less buoyant continental lithosphere.



A detailed description of the crust and lithospheric mantle thicknesses together with other information for each plate can be found in Table 2.

2.2. Strain-Induced LPO

The development of crystal aggregates LPO in the upper mantle depends on several deformation mechanisms, including plastic deformation, dynamic recrystallization (by sub-grain rotation and grain-boundary migration), and grain-boundary sliding (Kaminski et al., 2004). Here, we use a modified version of D-REX (Kaminski et al., 2004), that incorporates these deformation mechanisms to compute the LPO and accounts for the non-steady-state evolution of geodynamic systems (Faccenda & Capitanio, 2013).

Lagrangian particles representing upper and transition zone mantle mineral aggregates are regularly distributed throughout the computational domain (25 km reciprocal distance along the three directions, for a total of 209,664 aggregates). Each particle consists of 1,024 randomly oriented crystals forming an initially isotropic upper mantle with harzburgitic composition (70% olivine and 30% orthopyroxene modal abundance) and transition zone mantle with a more fertile, pyrolitic composition (60% spinel and 40% majoritic garnet) (Faccenda, 2014). These particles are then advected by means of the Eulerian velocity field obtained by the macro-flow modeling. At each time step, the orientations of these crystals change in response to gradients in the velocity field, generating LPO. We only compute the strain-induced fabrics in the upper mantle (from the Moho to the 410 km discontinuity) as SKS splitting parameters are mostly sensitive to this mantle layer (Sieminski et al., 2008). The upper mantle fabrics are reset and transformed to isotropic transition zone mantle aggregates upon crossing the Olivine-Spinel phase transition, and vice versa. Crystal aggregates of the transition zone are always assumed to be seismically isotropic (i.e., random crystal orientation). We use the same dimensionless crystallographic parameters as in Rappisi and Faccenda (2019) with the nucleation rate $\lambda^* = 5$, the grain-boundary-mobility $M^* = 1$ and the threshold volume fraction $\chi^* = 0.9$. When compared to previously published parameters calibrated with laboratory experiments (e.g., Boneh et al., 2015; Faccenda & Capitanio, 2013; Kaminski et al., 2004), the low M^* and high χ^* yield weaker fabrics and smaller amounts of seismic anisotropy that are more consistent with seismological observations.

2.3. SKS Splitting

The synthetic SKS splitting is calculated using routines included in the software package FSTRACK (Becker, 2006). The code first obtains a pulse seismogram via inverse Fourier transform using the method of Kennett (2009) with anisotropic extensions (Booth & Crampin, 1983; Chapman & Shearer, 1989) by computing the harmonic response of a horizontal layer stack to an incident plane wave (5° for typical SKS arrivals) over a range of frequencies (0–25 Hz). Synthetic seismograms in the SKS band (3.3–10 s) are then constructed applying band-pass filters from 0.1 to 0.3 Hz. Successively, the splitting is determined with the cross-correlation method of Menke and Levin (2003). A 2D grid of virtual seismic stations regularly spaced is constructed in the model. After recovering the elastic tensors from each aggregate's stiffness matrix, we build below each station and down to 400 km a vertical stack of horizontal layers (minimum thickness of 25 km) where the elastic tensor of each layer is radially averaged within a distance of 50 km. The SKS splitting parameters of each seismic station are obtained by averaging all the fast azimuths and delay times measured by rotating the vertical stack of elastic tensors by 5° intervals around the *y*-axis.

2.4. P-Wave Anisotropy and S-Wave Azimuthal Anisotropy

P-wave anisotropy is computed for each crystal aggregate as $Vp_{anis}(\%) = (Vp_{max} - Vp_{min})/(Vp_{max} + Vp_{min}) * 200$, where Vp_{max} and Vp_{min} are the max. and min. P-wave velocities evaluated for different incidence angles and azimuths. Another precious source of information is represented by the azimuthal anisotropy of Rayleigh waves, which however has a low lateral resolution. To roughly simulate this effect, we have interpolated the elastic tensors of crystal aggregates to a grid with node spacing of $100 \times 25 \times 100$ km. The fast azimuth ϕ and magnitude *G* of the S-wave azimuthal anisotropy are then computed as, respectively:

$$\phi = 0.5 \cdot \tan^{-1}(G_s/G_c) \tag{11}$$



Table 2

Rheological Parameters, Effective Thickness, and Age of the Different Plates for the Reference Model CM

Effective thickness					2
Plate name	(1	xm)	Friction coefficient (-)		Density (kg/m ³)
MODEL CM					
Lithosphere	Crust	Mantle	Initial	Final	
Eurasia (150 Myr)	20		0.020	0.005	$\rho_{mantle} - 400$
		60	0.600	0.400	$ ho_{mantle}$
Africa W (150 Myr)	40		0.020	0.005	$\rho_{mantle} - 400$
		60	0.600	0.400	$ ho_{mantle}$
Africa E (150 Myr)	25		0.020	0.005	$\rho_{mantle} - 400$
		65	0.600	0.400	$ ho_{mantle}$
Adria promontory (150	40		0.600	0.400	2700
Myr)		50	0.600	0.400	$ ho_{mantle}$
Adria thin margin (90	15		0.020	0.005	$\rho_{mantle} - 200$
Myr)		70	0.600	0.400	$ ho_{mantle}$
Ionian Ocean (90 Myr)	10		0.020	0.005	$ ho_{mantle}$
		80	0.600	0.400	$ ho_{mantle}$
Upper plate (1 Myr)	10		0.010	0.010	$ ho_{mantle}$
		80	0.600	0.400	$ ho_{mantle}$
Weakzone (70 Myr)	20	-	_	-	3200
MODEL A					
Adria thin margin is absent					
MODEL B					
Adria promontory is absent					
MODEL C-F1-F2					
Adria promontory and Adria thin margin are absent					
MODEL D1					
Adria Promontory crust	"	"	"	"	$\rho_{mantle} - 500$
MODEL D2					
Adria Promontory crust	22	"	33	33	$\rho_{mantle} - 400$
MODEL D3					
Adria Promontory crust	22	"	22	33	$\rho_{mantle} - 300$
MODEL D4					
Adria thin margin crust	22	**	55	2 2	$\rho_{mantle} - 400$
MODEL E					
Adria Promontory crust	"	"	0.020	0.005	"

Note. For all the other models, we only indicate the differences with respect to model CM. ρ_{mantle} is the mantle density from the thermodynamic databases. The viscosity of the weakzone is constant (10¹⁸ *Pas*). The employed low friction coefficients for crustal material ensure lubrication at the plates contact and one-sided, terrestrial-like subduction geometry (Gerya et al., 2008).



$$G = \sqrt{G_c^2 + G_s^2} \tag{12}$$

where $G_c = (C_{55} - C_{44})/2$, $G_s = C_{45}$ and C_{55} , C_{44} , C_{45} are the elastic tensor moduli.

3. Results

3.1. Reference Model CM

In this section, we describe in detail the tectonic evolution of the Reference Model CM. This model, among the several ones that have been tested as a function of the rheological parameters and of the plates structural/geometrical characteristics, is capable of reproducing the main geological and geophysical observations of the Central Mediterranean given the intrinsic model limitations. The influence of the imposed structural heterogeneities and rheological parameters are discussed in the next section.

The Reference Model CM is characterized by the presence of structural heterogeneities in the Adria continental plate, that is, a thinned margin in its northern portion and a stiffer promontory in its center as proposed by Lucente and Speranza (2001) and Lucente et al. (2006) (Figure 2). In this model, the subduction, driven solely by the slab negative buoyancy, initiates through progressive bending and roll-back of the oceanic plate, and causes a homogeneous stretching of the overriding lithosphere (Figure 3, Movie S1). The Ionian slab and trench rapidly migrate south-eastward with episodes of slab lateral tearing, segmentation, and break-off when a continental margin enters the trench.

Specifically, the western part of the trench reaches the African continent in ~4.5 Myr. When the trench impacts with the African margin, part of the continental lithosphere subducts and reaches the depth of ~150 km, and at ~6.4 Myr slab tearing occurs at the transition between the oceanic lithosphere and the more buoyant continental lithosphere (Figure 3a). Lithosphere tearing subsequently propagates eastward along the African passive margin, favoring slab rollback toward the east.

At \sim 7–8Myr (Figure 3b) the north-eastern portion of the trench progressively collides with the thin north-western margin of the Adria plate and the subsequent continental crust subduction induces (a) along-trench buoyancy variations resulting in a progressive curvature of the trench and (b) tearing of the slab north-eastern edge along the oceanic-continental lithosphere boundary (Figure 3c). After \sim 10 Myr, the trench reaches Central Adria with partial subduction of the stiffer continental promontory. At the same time, the subducting slab has already reached the lower boundary of the model and stagnates horizontally in the mantle transition zone.

Between ~15 and ~16 Myr, slab break-off occurs beneath the Adria promontory. The rupture rapidly propagates laterally, allowing, in only a few million years, the formation of a wide slab window that breaks the single arc geometry and creating two separated arcs (Figure 3d). The slab northern segment retreats eastward until it breaks at a depth between 150 and 200 km (~18 Myr), while the remaining Ionian slab becomes narrower and continues to migrate south-eastward.

After ~25–30 Myr (Figures 3e and 3f), slab remnants are found in model areas corresponding to the present-day northern Apennines and southern Tyrrhenian sea. The northern slab segment is hanging down to ~150 km depth and extends deeper from ~400 km down to about ~660 km depth. The southern slab segment instead extends continuously from the surface down to the mantle transition zone although incipient detachment is observed in the eastern African margin.

3.2. Comparison Between the Reference Model CM and Other Models

Here, we want to evaluate the influence on trench shape and on the occurrence and timing of slab tears (e.g., Mason et al., 2010) of the structural heterogeneities within the Adria plate. We explore a wide range of models (Table 2) where, with respect to the Reference Model CM, we varied the geometry of the Adria plate and the buoyancy and stiffness of the Adria promontory and Adria thin margin crust. Unfortunately, the past extent of the Adria plate and the geometry of its passive margins are quite uncertain, mostly because the latter are now subducted. As a consequence, in the literature there appears to be consensus on the Mid-





Figure 3. Snapshots of the Reference Model CM evolution. In light gray, the subducted slab (contour at T = 1,573K) below ~120 km depth. The Eurasian plate was opacified in the 1,600 < x < 2,600 km range for a better visualization of the subducted slab. The solid black line indicates the coastlines at 30 Ma (van Hinsbergen et al., 2014), while the dashed black line indicates the present-day coastlines of peninsular Italy.

to-Late Cenozoic kinematic evolution of the Ionian subduction margin, while few studies have attempted to reconstruct the distribution of oceanic and continental portions of the Adria plate, and thus the geometry of its margins. For this reason, here we have tested different potential margin geometries and plate configurations (homogeneous vs. heterogeneous plate, straight vs. curved passive margin), partly following those suggested in the literature (Faccenna et al., 2014; Lucente & Speranza, 2001; van Hinsbergen et al., 2014), and partly hypothesizing some potential geometries like those depicted in Figure 6.

As expected, the initial subduction dynamics of all models are similar to that of Model CM: the initial slab sinks through the mantle and retreats to the south-east. When the trench impacts with the African margin, lateral slab tearing favors subsequent eastward rollback. Important differences among models occur when the trench reaches the Adria continental margin.





Figure 4. Snapshot at ~30 Myr of: (a) Model CM, (b) Model A, (c) Model B, (d) Model C, (e) Model F1, and (f) Model F2. Color-coding and coastlines as in Figure 3.

When the thin continental margin in the northern Adria plate is absent (Model A), the eastward migration of the northern Ionian trench is hindered and slab tearing below the Adria promontory occurs ~3 Myr later than in the Model CM (~19 Myr). Over the next few millions of years, the tear spreads laterally forming a wide slab window (Figure S1b) and leading to complete slab detachment at ~30 Myr (Figure 4b) with slab remnants down to ~350 km depth. This results in a lower amount of retreat of the northern Ionian trench, whose final shape appears linear and not arcuate as in Model CM (Figures 5a, 5b, S2b and S2c).

When the Adria promontory is not included (Model B), the central slab window and the consequent double arc geometry are not generated. At ~20 Myr the northern part of the trench is curved and has reached the same position as Model CM (Figures 5a and 5c). The Ionian slab totally breaks off between ~24 and ~25 Myr (Figure S1c), resulting in a smaller amount of trench retreat to the south when compared to Model CM (Figures 5a, 5c, S2b and S2d). At ~30 Myr, we find remains of hanging subducted lithosphere only in the central part of the Adria plate (Figure 4c).





Figure 5. Top view of the trench position evolution up to 30 Myr for: (a) Model CM, (b) Model A, (c) Model B, (d) Model C, (e) Model D1, (f) Model D4, (g) Model E, (h) Model F1, and (i) Model F2. The color bar indicates the time in million years. In models CM, B, D1, D4, E, and F2, the northern section of the trench retreats till the Adriatic Sea in ~30 Myr. The retreat of the Ionian slab till the present-day position in the Calabro-Peloritan region is reproduced in models CM, D1, and D4. Coastlines as in Figure 2.

When both the stiff continental promontory and the thin continental margin are removed (Model C), a tear occurs at ~21 Myr in the slab central portion and again quickly spreads toward its edges. At ~23 Myr, the southern edge of the slab breaks at a depth between 150 and 200 km (Figure S1d), while the remaining northern slab breaks at a depth of ~300 km at ~30 Myr (Figure 4d). The final position of the trench is equal to Model A in the northern part and equal to Model B in the southern part (Figures 5b–5d and S2c–S2e).

The continental crust is much more buoyant with a density of ~2,700–3,000 kg/m³, compared to 3,300 kg/m³ for the mantle. Here, we further test the influence of the density of the promontory continental crust in models D1, D2, and D3, which is 500, 400, and 300 kg/m³, respectively, lighter than the mantle. This density is higher than that in Model CM where it was set constant and equal to 2,700 kg/m³. The models evolution remains very similar to Model CM, and a wide slab window develops leading to the formation





Figure 6. Top view of the initial model setup for the models with a homogeneous Adria plate (without structural heterogeneities): (a) Model F1 and (b) Model F2). Coastlines as in Figure 2.



of two separate arcs after the subduction of the denser continental promontory. At ~30 Myr the shape and position of the two trenches of these three models are very similar to Model CM (Figures 5a and 5e). An analogous result was obtained for Model D4, where a more buoyant Adria thin continental margin is tested by decreasing the continental crust by 200 kg/m³ (i.e., $\rho_{crust} = \rho_{mantle} - 400 \text{ kg/m}^3$) relative to that of Model CM (Figures 5a and 5f). In general, the influence of continental crust density of different Adria plate domains is minor and exerts negligible effects on the final model evolution.

In Model E, we have instead inserted a less rigid promontory, whose initial and final friction coefficients are set equal to those of the Adria plate (0.020 and 0.005). Although the overall evolution of the subduction is similar to that of the Model CM, the timing of slab break-off is different. At ~25 Myr, total slab detachment has already occurred in both the northern and southern slabs, that is, at least 5 million years earlier than the Model CM. This causes a minor retreat of the southern trench (Figures 5a, 5g, S2b and S2f).

Finally, we tested two models (models F1 and F2) with a homogeneous Adria plate (without lateral heterogeneities). In these models, the initial geometry of the Adria plate is simpler than in Model CM (Figure 6). Model F1 shows a westward convex geometry of the Adria plate western margin (Figure 6a), which is instead straight in Model F2 (Figure 6b). In Model F1, the evolution of the Ionian slab in the mantle is well reproduced (Movie S2), with the formation of the wide slab window under the central part of Adria (Figure S1e). At ~30 Myr the northern slab is broken at a depth of about 200 km while the southern slab is still continuous (Figure 4e). Compared to the Model CM, however, the retreat of the two trenches is smaller. Furthermore, the trench arcuate shape observed in Model CM is not recreated (Figures 5a, 5h, S2b and S2g). The evolution of Model F2 is instead similar to that of Model B. When the Adria plate passive margin enters the trench, the slab breaks off completely without developing the slab window and the two separate arcs (Figure S1f). At ~30 Myr, we find remains of subducted lithosphere only in the central part of the Adria plate (Figure 4f). Also, in this case, the final shape of the trench is linear (Figures 5i and S2h).

3.3. Upper Mantle Flow, LPO, and Synthetic Seismic Anisotropy

The mantle flow induced by slab subduction for Model CM is characterized by the presence of poloidal and toroidal components (Movie S1). The initial slab subduction is associated with a dominant poloidal asthenospheric flow and mantle upwelling in the mantle wedge. At the same time, slab rollback is accommodated by return flow from the sub-slab region toward the mantle wedge around the slab edges and, at later stages, through the newly formed slab window.

The evolution of the subduction margin produces strong upper mantle fabrics in the area surrounding the retreating slab to a depth of 410 km and considerable amount of seismic anisotropy. In the supra-slab upper mantle, the general trend of the fast azimuths is trench-perpendicular, while the sub-slab mantle is characterized by trench-parallel extension and anisotropy (Faccenda & Capitanio, 2012, 2013) (Figures 7, 8 and S2). In areas characterized by slab break-off/lateral tearing and vertical flow such as around the modeled Northern Apennines and Calabrian arc, the shear wave anisotropy is relatively small due to the dipping fast axes of olivine crystals at shallow mantle depths (to which SKS phases are mostly sensitive). Synthetic SKS splitting measurements (Figures 7 and S2b) indicate that the fast shear wave component orients perpendicular to the past trench positions in the back-arc region. This results in a smoothly rotating direction pattern, with a dominance of NW directions in the north turning progressively toward EW. The back-arc region is characterized by very high delay times ($\delta t = 2 - 3.4 s$), which reduce closer to the trench ($\delta t < 1.5 s$) and orient parallel to it in the fore arc region. Around the slab edges, the teleseismic fast shear wave component traces the underlying return flow ($\delta t = 1 - 2 s$) with a circular pattern surrounding the southern and the norther trenches. The P-wave and Rayleigh wave fast azimuths are consistent with those of the SKS-splitting data (Figure 8). Around the region corresponding to the northern Apennines there is significant change in P-wave and SKS anisotropic patterns between ~ 20 and ~ 30 Myr (Figure 9). At earlier stages, strong and trench-parallel P-wave and SKS anisotropy results from the retreating western Adria margin (Figures 9c and 9e). At later stages, however, the trench-parallel SKS fast azimuths are substantially reduced and the P-wave fast directions at 130 km depth turn to trench-perpendicular dipping toward the sinking and detached Ionian slab (Figures 9d and 9f). This is because the detached Ionian slab induces a downward and radially converging mass flux (Figures 9a and 9b) in the shallow upper mantle that progressively re-orients





Figure 7. (a) SKS-splitting measurements in the Central-Western Mediterranean (Becker et al., 2012) color-coded by the error in the orientation compared with synthetic SKS splitting measurements for Model CM at \sim 20 My (green bars). The EW green bar in the upper left corner indicates 2 s. (b) Split time error and split azimuth error histograms.

the fast axis of the olivine crystals from ~NW to ~NE (Figures 9g and 9h; Vp pole figures). The change in fast axis dip from sub-horizontal to SW dipping is responsible for the decrease in SKS splitting δt (Figures 9g and 9h; S-wave anisotropy pole figures).

The mantle flow evolution of the other models (A, B, C, D1, D2, D3, D4, E, F1 [Movie S2], and F2) is similar to Model CM, where lithospheric deformation and asthenospheric flow are controlled by the slab roll-back and poloidal and toroidal components of mantle circulation are both active. An important difference occurs in the models where a shallow slab window does not form (B, C, F1, and F2). In this case, the return flow





Figure 8. Top-view of the Reference Model CM at ~30 Myr showing (a) Vp_{anis} for selected aggregates at a depth of 150 km, with the length and color of each bar proportional to the amount of anisotropy and the direction indicating the direction of Vp_{max} ; (b) The S-wave azimuthal anisotropy at a depth of 140 km with the orientation and length of the blue bars representing the fast azimuth Φ and magnitude *G* of azimuthal anisotropy, respectively. The EW blue bar outside the domain indicates 1% of azimuthal anisotropy. The gray surface shows the material with P-wave (left) and S-wave (right) isotropic velocity anomaly 1% with respect to the depth averages velocities below 140 km depth, and corresponds to the Ionian slab. The coastlines of Africa at 30 Ma and the present-day coastlines of peninsular Italy are indicated in black for reference.

entering the window is not established and a larger toroidal flow component around the edges of the wider slab is observed.

We also calculated the synthetic SKS splitting parameters for models A, B, C, E, F1, and F2 (Figures S2c–S2h). In the bark-arc region, SKS splitting patterns of these models are quite similar to those of Model CM, with a fast shear wave component aligned trench-perpendicular and showing an anticlockwise rotating pattern. Substantial differences are found in regions near the trench affected by the slab break-off and lateral tearing. In the models A, B, C, E, F1, and F2, the clockwise pattern of the fast azimuths surrounding the southern trench is characterized by shorter delay times ($\delta t < 1 s$) due to a lower amount of trench retreat (Figures S2c–S2h, 5b–5d, and 5g–5i). Conversely, in models A, C, F1, and F2 (Figures S2c, S2e, S2g, and S2h), the circular pattern surrounding the northern trench is well highlighted by the higher delay times ($\delta t = 1 - 1.5 s$). This reflects the strong toroidal flow establishing around the rectilinear slab present at shallow depths along the entire Adria plate. Furthermore, in models B, C, F1, and F2 (Figures S2d, S2e, S2g, and S2h), differences emerge in the Adria central portion where the fast axes orient parallel to the trench as the absence of the slab window does not produce any return flow through it.

4. Discussion

4.1. Data and Model Limitations

The direct application of numerical modeling to natural cases is affected by the modeling assumptions and limitations. For example, in our large-scale geodynamic models, the relative position of the plates slightly differs from the present-day one. This is because we used an initial configuration defined at ~30 Ma according to the paleogeographic and tectonic reconstructions from Lucente and Speranza (2001), Lucente et al. (2006), Faccenna et al. (2014), and van Hinsbergen et al. (2014), and we have not applied a convergence rate between the plates (self-consistent subduction). As such, the African continent never reaches its current position. It is worth noting though that we mainly focus on the subduction system and its effect on mantle flow and anisotropy patterns. An initial geometry defined at ~30 Ma should not have a strong impact on mantle flow directions and splitting parameters since the slow Africa-Europe plates convergence has not caused a drastic change in plates position over this time span.



Journal of Geophysical Research: Solid Earth



Figure 9.



The model domain is defined in Cartesian coordinates, which could cause discrepancies when comparing our synthetic observations with the real measurements (e.g., Figure 7). Furthermore, tectonic reconstructions show that at ~30 Ma, an incipient slab (150 km), located from Gibraltar to Corsica, was already subducted in the upper mantle (Faccenna et al., 2014; van Hinsbergen et al., 2014). However, to trigger a "spontaneous" subduction system, the trench has been initially positioned further south with the slab extending to a depth of 300 km to model a more developed subduction and increase the slab negative buoyancy. This could cause a difference in rates of Ionian slab retreat at the model early stage when compared to those reported in the literature.

The Central Mediterranean dynamics has been likely affected by the presence of multiple subducting slabs rather than a single one as in our models. Numerical and analog models with multiple subductions demonstrated that the number and geometry of interacting slabs strongly affect the overall force balance, and that interactions through the mantle can exert a primary control on the geometry and kinematics of subduction (Holt et al., 2017, 2018; Jagoutz et al., 2015; Luth et al., 2013; Peral et al., 2020). In addition, it remains difficult if not impossible to evaluate the role of other structural heterogeneities (i.e., oceanic transform faults) and geometrical complexities that might have been potentially present within the subducted Ionian plate and eastern Adria margin and that have not been considered here.

It is important to point out that validation of geodynamic models through comparison of the seismological synthetics with observations is limited by (a) uncertainties concerning the seismological data sets and related to the fact that, for example, a good azimuthally coverage is needed to properly describe the back-azimuthal dependency of the SKS splitting parameters, (b) lack of anisotropy data in oceanic regions and certain continental areas (i.e., Central Apennines), and (c) ignoring other sources of seismic anisotropy in our models. The latter include fossil LPO fabrics in the crust and mantle of the subducted and overriding lithosphere, and preferentially aligned compositional heterogeneities and fluid-filled cracks. About crustal fabrics, we note that SKS and SKKS waves have a wide lateral sensitivity (the Fresnel zone is typically on the order to 100 km or more in the upper mantle), and the role of the few tens of km thick subducted continental crust on seismic anisotropy of long-wavelength waves is likely minor. Little is known about the mantle fossil anisotropy which might be present in the Ionian-Adria lithosphere, but we note that it would be quite a coincidence the situation where the orientation of the rotated (i.e., subducted) fabrics is consistent with the systematic along-strike orientation of the SKS fast azimuths along the mountain belts with a very complex arcuate geometry. As an example, Petrescu et al. (2020) carried out Fresnel analysis of the SKS splitting data with trench-parallel patterns along Alps and placed the source of anisotropy at asthenospheric depths. The anisotropy produced by fluid-filled, 10 km thick crustal faults in the Apennines is typically small (<0.1 s in average; Baccheschi et al., 2016), while Faccenda et al. (2019) have demonstrated that extrinsic anisotropy due compositional heterogeneities is negligible. Hence, we believe that strain-induced LPO fabrics in the recently deforming hot mantle should explain most of the observations in the area.

Modeling of strain-induced LPO in upper mantle rocks has also some limitations related to the incomplete knowledge of the relative contribution of the different creep mechanisms as a function of the local P-T conditions, and to the employed numerical approach (for a recent review of this topic, see Hansen et al., 2021).

Despite these limitations, we find a good correspondence between the modeled and observed surface and deep isotropic structures, and seismic anisotropy patterns. This might indicate that our numerical geodynamic models are able to capture the first order of the overall evolution and the current geological scenario of the Central Mediterranean. In particular, the models reproduce several episodes of slab lateral tearing and break-off that have been proposed according to geological and seismological data.

Figure 9. Compositional field and seismic anisotropy in the reference Model CM at ~20 Myr (left column) and ~30 Myr (right column). First row is the compositional field cropped around the region corresponding to the Apenninic belt with a vertical slice of the velocity field crossing the Apennines. The arrows length is proportional to the velocity magnitude, and are color-coded according to the velocity vertical (*Y*) component. Second row is the P-wave anisotropy at 137.5 km depth. The bars length is proportional to the Vp_{anis} magnitude, and are color-coded according to the Vp_{anis} vertical (*Y*) component. Third row is the SKS splitting. The transparent blue and red volumes denote the lithosphere deeper (lighter) or shallower (darker) than 200 km depth. Fourth row shows the pole projections P-wave velocity (m / s) and $Vs_{anis} = (Vs_1 - Vs_2) / (Vs_1 + Vs_2) * 200$ as a function of the propagation direction for a mantle crystal aggregates representative of the northern Apennines area (X = 837.5 km, Y = 137.5 km, and Z = 1337.5 km; shown by the circled cross in panels (c-f)). The bars in the Vs_{anis} polot indicate the polarization direction of the Vs_1 component. The black square and white circle indicate max. and min. values, respectively.



4.2. Comparison With Seismic Tomography Models

A common feature of all of our synthetic models is a high-velocity body (Figure 8) arranged horizontally over the 660 km discontinuity. This high-velocity anomaly is a prominent feature in P-wave tomographic models (Amato et al., 1993; Lucente et al., 1999; Piromallo & Morelli, 2003; Selvaggi & Chiarabba, 1995; Spakman et al., 1993; Spakman & Wortel, 2004; Van der Meer et al., 2018) and it has been interpreted as the Ionian slab lying and broadening at the base of the upper mantle.

The slab remnants found in the later stages of Model CM under the regions corresponding to the Northern Apennines and Calabrian arc are consistent with (a) the observed intermediate and deep seismicity (Chiarabba et al., 2005), and (b) the two pronounced high-speed surface anomalies separated by a slow velocity anomaly (Central Apennines Slab Gap) detected by seismic tomography models. A high-velocity anomaly is detected at 250 km and deeper below the entire Apenninic belt that can be related to subducted Ionian oceanic lithosphere. The portion of the slab under the northern Apennines extends down to 150 km depth and from ~200 km down to about 660 km depth (Figures 3d and 9a). This is consistent with recent tomographic studies (El-Sharkawy et al., 2020; Giacomuzzi et al., 2012; Spakman & Wortel, 2004) suggesting that the Northern Apennines slab reaches a depth of only 150-300 km. The vertical width of the slab gap increases with time, such that top portion of the detached slab is at 400 km depth at ~30 Myr (Figures 3f, 4a and 9b). The northern section of the modeled trench retreats westward and already at ~20 Myr resembles the arcuate shape of the present-day northern Apennines extending from the western Po Plain to the Adriatic Sea. Toward the south, the modeled Ionian slab reaches its present-day position in the Calabro-Peloritan region again at ~20 Myr (Figure 5a). According to Model CM (Figures 3f and 4a), the Calabrian slab is continuous from the surface up to a depth of 660 km beneath the northern and central Calabria but the detachment has already occurred beneath north eastern Sicily (El-Sharkawy et al., 2020; Giacomuzzi et al., 2012; Neri et al., 2012; Presti et al., 2019; Scarfi et al., 2018). Thus, the modeled structures of Model CM at ~20 Myr appear to be quite compatible with the present-day surface and deeper configuration (Figure 9, left column). This is in agreement with the fact that our initial model geometry characterized by a deeper slab than what reconstructed at ~30 Ma by Faccenna et al. (2014) and van Hinsbergen et al. (2014) is likely representative of a more a recent stage of the Ionian subduction.

A substantially similar evolution was obtained for the models A, D1, D2, D3, D4, E, and F1 (Figures 4b and 4e), although with a minor retreat of the trenches in the models A, E, and F1 (Figures 5b, 5g, 5h, S2c, S2f, and S2g). This implies that continental crust thickness and strength, rather than its density contrast with respect to the mantle (models D1, D2, D3, and D4), play a major role in defining the modeled final configuration.

Results inconsistent with tomography observations were instead obtained for the models B, C, and F2, where the absence of the stiffer Adria promontory causes the complete break-off of the Ionian slab (Figures 4c, 4d, and 4f). In Model B, we find hanging subducted lithosphere only below the central Apennines, which is in contrast with seismic tomographies supporting the presence of a slab gap (Figure 4c). In models C and F2, a hanging slab is found under the entire Apenninic chain (Figures 4d and 4f). Moreover, the absence of a thin continental margin in the northern Adria plate (models A, C, F1, and F2) hinders the western retreat of the northern trench and the formation of an arcuated trench similar to that observed along the Apenninic chain (Figures 5b, 5d, 5h, 5i, S2c, S2e, S2g, and S2h). It is important to mention that including a free surface condition near the top boundary could affect the final geometry of the model in the continental areas, as portions of the continental crust entering the trench could be thrusted over the foreland, shifting the trench more eastward (Faccenda et al., 2009).

In synthesis, our geodynamics experiments highlighted that the initial subduction dynamics of all models are similar and important differences in terms of rollback rates, trench shape, and occurrence and timing of slab tears occur in the final stages of their evolution when the Adria continental margin enters the trench. In particular, our results show that:

1. The presence of a thin continental lithosphere (models CM, B, D1, D2, D3, D4, and E) favors the curvature of the Northern Apennines trench and its retreat till the Adriatic Sea, which suggests that the arched shape of the northern Apennines is a consequence of deep processes occurring inside the Earth, that is, the lateral flexion of the subducting Adriatic plate (Lucente & Speranza, 2001);



- 2. The development of a slab window below the Central Apennines and double arcs geometry, as shown by seismic tomographies, occurs in the models where a stiffer continental promontory in central Italy is inserted (models CM, A, D1, D2, D3, D4, and E) and/or the Thyrrenian passive margin of the Adria plate is curved (Model F1);
- 3. The presence of both structural heterogeneities within the Adria plate (promontory and thin margin) helps the retreat of the Ionian slab till the present-day position in the Calabro-Peloritan region.

4.3. Comparison With Observed Seismic Anisotropy

Figure 7 shows a comparison between predicted and observed (by Becker et al., 2012) SKS splitting measurements in terms of azimuth and time errors. More in detail, panel "a" shows the observed measurements color-coded by the error in the orientation compared with synthetic predicted from Model CM at ~20 My (green bars). The general pattern of the synthetic 3D anisotropy calculations matches the observed data in the whole study region with an average misfit angle of about 26° , but lower values are found in the Calabrian arc (0–10°), southern France (0–20°) and eastern Alps (0–10°), although the mantle flow in the latter area could be affected by its proximity to the model edges. In contrast, a significant mismatch between predicted and observed fast azimuths is found in the Dinaric Alps and Central Apennines where the predictions mainly exhibit a trench-parallel preferential direction with an average angle rotated by 90° with respect to the observations. These discrepancies could arise from the fact that in this study we have ignored the presence of the Alpine and Dinaric slabs. In fact, the Adria microplate is subducting on both the western and eastern margins, which may cause a strong interaction between the two subduction zones (Király et al., 2018) that has not been considered in our models. The comparison between observed and predicted data is also limited by the fact that some areas are not covered by real measurements as in the case of the Tyrrhenian Sea, the Ionian Sea, and partly in the Central Apennines.

Bearing in mind these discrepancies, our results (Figures 7–9 and S2) reveal the dominant role of subduction-induced mantle flow in generating the observed seismic anisotropy and shear wave splitting. Moving from West to East, three main areas characterized by different anisotropy patterns are identified.

Around the area corresponding to the Liguro-Provençal basin the synthetic SKS splitting, P-wave anisotropy, and Rayleigh wave azimuthal anisotropy measurements exhibit a smooth $\sim 90^{\circ}$ rotation, with a dominance of NW orientation in the north (southern France) turning progressively toward EW in the Corsica-Sardinia block. We confirm the interpretation proposed by Lucente et al. (2006) that relates this feature to the formation of the Liguro-Provençal basin and to the clockwise rotation of the Corsica-Sardinia block.

In the modeled Tyrrhenian domain, the fast shear wave component orients parallel to the stretching direction of the back-arc basin, which is consistent with the observations (Jolivet et al., 2009; Lucente et al., 2006). However, delay times are higher ($\delta t = 2 - 3.4$ s) due to the thick and homogeneous anisotropic layer with sub-horizontal fast axis extending down to the base of the upper mantle. This might indicate that either dislocation creep is active down to only 200–250 km (Karato & Wu, 1993) or that mantle fabrics are weaker than those modeled.

Beneath the Apennines, the pattern of observed SKS fast axes abruptly changes orientation and becomes parallel to the mountain chain, following its curvature from NW-SE in the north, to a more NS orientation along the Apennines foreland and to NE-SW in the farther south Calabria. Along the northern Apennines (Figure 9), the trench-perpendicular P-wave anisotropy measured by Hua et al. (2017) could be interpreted with the presence of a large gap in the sinking Ionian slab as in the latest stages of Model CM (Figures 9b and 9d). At earlier stages of Model CM, instead, that is, during or right after slab detachment, the SKS-splitting delay times are more consistent with the observations, but the P-wave fast azimuths are trench-parallel (Figures 9a, 9c and 9e). Because of the striking difference between these two patterns, it follows that a more detailed characterization of the seismic anisotropy in the northern Apennines could provide useful information to determine whether the slab is continuous or not in the upper mantle. In the Sicily channel, smaller-scale features show trench-perpendicular SKS azimuths at the western corner of the Calabrian slab, where SKS fast axes turn to NS and then EW in the Tyrrhenian Sea. According to our models, these patterns reflect the retreat of the Adria-Ionian trenches and related trench-parallel extension in the sub-slab mantle, and toroidal flow around the Calabria slab (Civello & Margheriti, 2004; Faccenna et al., 2007, 2014;



Jolivet et al., 2009). In our reference Model CM, the SKS fast azimuths are somewhat trench-perpendicular beneath Central Adria as a result of the mantle flowing through the slab window. However, the scarcity of splitting data in this region does not allow to safely infer the mantle flow at depth and compare our results with the few available observations.

5. Conclusions and Outlook

We combined macro-scale geodynamic modeling with micro-scale simulations of strain-induced upper mantle fabrics and seismological synthetics to reproduce and constrain the tectonic evolution of the Central Mediterranean over the last $\sim 20-30$ Myr. Starting from different tectonic scenarios, we have employed 3D thermo-mechanical simulations to test the role of lithospheric structural heterogeneities and certain rheological parameters on the self-consistent model evolution. After calculating the upper mantle elastic properties as a function of the strain history and local P-T conditions, the model results were then validated by comparing seismological synthetics (isotropic P-wave anomalies, P-wave anisotropy, SKS splitting, and Rayleigh azimuthal anisotropy) and major tectonic features (i.e., slab and trench geometry) with observations.

Our modeling results bring new insights into the complex structure of the upper mantle beneath the region confirming that the main geological and geophysical observables in the Central Mediterranean can be directly linked to the recent dynamics of the Ionian slab and Adria plate. Furthermore, our results suggest that lateral variations in lithosphere thickness and stiffness can substantially influence the tectonic history of a natural environment. We show that the presence of structural heterogeneities within the Adria plate as imaged by, for example, Miller and Piana Agostinetti (2012) and Panza et al. (2003), and/or the geometry of its Tyrrhenian passive margin plays a fundamental role on:

- 1. The curvature of the Northern Apenninic trench and its retreat to the Adriatic Sea. This is favored by the presence of the thin continental lithosphere in the northern Adria plate that is prone to subduction.
- 2. The development of a slab window below the Central Apennines separating two high-velocity anomalies as shown by seismic tomographies. A good fit between predicted and observed slab morphology was obtained for the models where a thick continental promontory lithosphere is inserted inside the Adria plate. In these scenarios, slab break-off occurs beneath the area corresponding to the Central Apennines and is favored when the thick promontory is bounded by lithospheric portions (the thin continental margin in the North and the oceanic Ionian plate in the South) that tend to subduct and retreat eastward. This rupture rapidly propagates laterally, allowing, in only a few million years, the formation of two separated arcs, the northern Apennines and the Calabrian arcs. Results inconsistent with tomography observations were instead obtained for the models where the absence of the stiffer Adria promontory results in the complete break-off of the Ionian slab.
- 3. The retreat of the Ionian slab till the present-day position in the Calabro-Peloritan region in models when both structural heterogeneities were present.

Despite some discrepancies due to the imposed model geometry and some limitations in the methods, the good correlation between predicted and observed slab morphology and seismic anisotropy patterns pose new important constraints on the recent evolution of the study area. Our geodynamic models explain the rotation of the Sardinian-Corsican block, the opening of the Liguro-Provençal and Tyrrhenian basins, the retreat of the Apennines and Ionic trenches until their present-day position. In particular, this work offers valuable constraints on the hypothesized break-off of the slab beneath the Central Apennines widely debated in the literature, giving a combined geodynamic and seismological reading. Our study demonstrates that this type of combined simulation is able to capture a first order of the overall evolution and the current geological situation of the region providing an overview on the processes that have led to the opening of the Central Mediterranean.

To obtain a better comparison between the synthetic and observed data and to limit the model uncertainties, we envisage that future numerical studies should attempt to (a) improve the model geometry by considering the Earth's sphericity and the Africa-Eurasia plates convergence, (b) account for the presence of the Alpine and Dinaric slabs that likely affect the mantle flow below the Adria plate surrounding regions, and (c) perform further seismological synthetics, such as P-wave anisotropic tomographic inversions (e.g., VanderBeek & Faccenda, 2021). At the same time, more abundant and high-quality seismological measurements could



provide more constraints on the upper mantle structure and flow dynamics. For example, at present, it is not yet clear whether the northern Apenninic slab is continuous down to the transition zone or not. Furthermore, SKS splitting in some specific areas such as the Central-Southern Apennines are sparse, such that there it is not possible to infer whether the shallow upper mantle is flowing through the Central Apennines Slab Gap as a result of the recent retreat of the nearby Calabrian and Northern Apennines slabs. Similarly, P-wave and S-wave anisotropic tomography models of the entire Mediterranean region would likely help in better constraining the recent dynamics and deep structure of this region. However, a single P-wave anisotropic tomography for the Alpine region (Hua et al., 2017) has been produced so far Hua et al. (2017), which is unexpected given the good azimuthal coverage of the area (e.g., Piromallo & Morelli, 2003).

Data Availability Statement

T. Gerya provided the I3MG code used for the subduction modeling. This code is available on Figshare (https://figshare.com/articles/software/I3MG/16707370). B. P. VanderBeek created the map in Figure 1b using the software GMT. The modified version of the D-Rex code used for the fabric modeling, and the routines used to calculate SKS splitting parameters, P-wave and Rayleigh wave anisotropy can be found inside the ECOMAN software package (https://newtonproject.geoscienze.unipd.it/ecoman/). The MATLAB toolbox geomIO used to define the geometry of the model initial setup can be found at https://geomio.bitbucket. io/. Paraview was used for graphic visualization of the model output (https://www.paraview.org/). Files for visualization in Paraview of the reference Model CM are available on Figshare (https://figshare.com/articles/dataset/File_for_visualization_in_Paraview/16438797). The map in Figure 1b is created with GMT 5.4.3 which is under a GNU Lesser General Public License. Data from real SKS splitting measurements were taken from (Becker, 2006) (https://www-udc.ig.utexas.edu/external/becker/sksdata). Reviews by Simone Pilia and Luca De Siena have substantially improved an earlier version of this article.

Acknowledgments

This study is supported by the ERC StG 758199 NEWTON. Open Access Funding provided by Universita degli Studi di Padova within the CRUI-CARE Agreement.

References

- Amato, A., Alessandrini, B., Cimini, G., Frepoli, A., & Selvaggi, G. (1993). Active and remnant subducted slabs beneath Italy: Evidence from seismic tomography and seismicity. *Annals of Geophysics*, *36*(2). https://doi.org/10.4401/ag-4272
- Arab, M., Maherssi, C. E., Granjeon, D., Roure, F., Déverchère, J., Cuilhé, L., et al. (2020). On the origin and consequences of crustal-scale extension between Africa and Sicily since Late Miocene: Insights from the Kaboudia area, Western Pelagian Sea. *Tectonophysics*, 795, 228565. https://doi.org/10.1016/j.tecto.2020.228565
- Baccheschi, P., Margheriti, L., & Steckler, M. (2007). Seismic anisotropy reveals focused mantle flow around the Calabrian slab (southern Italy). *Geophysical Research Letters*, 34(5). https://doi.org/10.1029/2006gl028899
- Baccheschi, P., Pastori, M., Margheriti, L., & Piccinini, D. (2016). Shear wave splitting of the 2009 L'Aquila seismic sequence: Fluid saturated microcracks and crustal fractures in the Abruzzi region (central Apennines, Italy). *Geophysical Journal International*, 204(3), 1531–1549. https://doi.org/10.1093/gji/ggv536
- Bauville, A., & Baumann, T. S. (2019). geomIO: An open-source MATLAB toolbox to create the initial configuration of 2-D/3-D thermo-mechanical simulations from 2-D vector drawings. *Geochemistry, Geophysics, Geosystems, 20*(3), 1665–1675. https://doi. org/10.1029/2018gc008057
- Becker, T. W. (2006). On the effect of temperature and strain-rate dependent viscosity on global mantle flow, net rotation, and plate-driving forces. *Geophysical Journal International*, 167(2), 943–957. https://doi.org/10.1111/j.1365-246x.2006.03172.x
- Becker, T. W., Lebedev, S., & Long, M. (2012). On the relationship between azimuthal anisotropy from shear wave splitting and surface wave tomography. *Journal of Geophysical Research*, 117(B1). https://doi.org/10.1029/2011jb008705
- Benoit, M. H., Torpey, M., Liszewski, K., Levin, V., & Park, J. (2011). P and S wave upper mantle seismic velocity structure beneath the northern Apennines: New evidence for the end of subduction. *Geochemistry, Geophysics, Geosystems, 12*(6). https://doi.org/10.1029/2010gc003428
- Bezada, M., Humphreys, E., Toomey, D., Harnafi, M., Dávila, J., & Gallart, J. (2013). Evidence for slab rollback in westernmost Mediterranean from improved upper mantle imaging. *Earth and Planetary Science Letters*, 368, 51–60. https://doi.org/10.1016/j.epsl.2013.02.024
- Boneh, Y., Morales, L. F., Kaminski, E., & Skemer, P. (2015). Modeling olivine CPO evolution with complex deformation histories: Implications for the interpretation of seismic anisotropy in the mantle. *Geochemistry, Geophysics, Geosystems, 16*(10), 3436–3455. https://doi. org/10.1002/2015gc005964
- Booth, D. C., & Crampin, S. (1983). The anisotropic reflectivity technique: Theory. *Geophysical Journal International*, 72(3), 755–766. https://doi.org/10.1111/j.1365-246x.1983.tb02831.x
- Buontempo, L., Bokelmann, G., Barruol, G., & Morales, J. (2008). Seismic anisotropy beneath southern Iberia from SKS splitting. *Earth and Planetary Science Letters*, 273(3–4), 237–250. https://doi.org/10.1016/j.epsl.2008.06.024
- Calcagnile, G., & Panza, G. (1980). The main characteristics of the lithosphere-asthenosphere system in Italy and surrounding regions. *Pure and Applied Geophysics*, 119(4), 865–879.
- Carminati, E., Lustrino, M., & Doglioni, C. (2012). Geodynamic evolution of the central and western Mediterranean: Tectonics vs. igneous petrology constraints. *Tectonophysics*, 579, 173–192. https://doi.org/10.1016/j.tecto.2012.01.026
- Chapman, C., & Shearer, P. (1989). Ray tracing in azimuthally anisotropic media—II. Quasi-shear wave coupling. *Geophysical Journal International*, *96*(1), 65–83. https://doi.org/10.1111/j.1365-246x.1989.tb05251.x



- Chiarabba, C., Jovane, L., & DiStefano, R. (2005). A new view of Italian seismicity using 20 years of instrumental recordings. *Tectonophysics*, 395(3-4), 251–268. https://doi.org/10.1016/j.tecto.2004.09.013
- Christensen, U., & Yuen, D. (1985). Layered convection induced by phase transitions. Journal of Geophysical Research: Space Physics, 90(B12), 10291–10300. https://doi.org/10.1029/jb090ib12p10291
- Civello, S., & Margheriti, L. (2004). Toroidal mantle flow around the Calabrian slab (Italy) from SKS splitting. *Geophysical Research Letters*, 31(10). https://doi.org/10.1029/2004gl019607
- Confal, J. M., Faccenda, M., Eken, T., & Taymaz, T. (2018). Numerical simulation of 3-D mantle flow evolution in subduction zone environments in relation to seismic anisotropy beneath the eastern Mediterranean region. *Earth and Planetary Science Letters*, 497, 50–61. https://doi.org/10.1016/j.epsl.2018.06.005
- Connolly, J. A. (2005). Computation of phase equilibria by linear programming: A tool for geodynamic modeling and its application to subduction zone decarbonation. *Earth and Planetary Science Letters*, 236(1-2), 524–541. https://doi.org/10.1016/j.epsl.2005.04.033
- Dewey, J., Helman, M., Knott, S., Turco, E., & Hutton, D. (1989). Kinematics of the western Mediterranean. Geological Society, London, Special Publications, 45(1), 265–283. https://doi.org/10.1144/gsl.sp.1989.045.01.15
- El-Sharkawy, A., Meier, T., Lebedev, S., Behrmann, J. H., Hamada, M., Cristiano, L., & Köhn, D. (2020). The slab puzzle of the alpine-Mediterranean region: Insights from a new, high-resolution, shear wave velocity model of the upper mantle. *Geochemistry, Geophysics, Geosystems*, 21(8), e2020GC008993.
- Faccenda, M. (2014). Mid mantle seismic anisotropy around subduction zones. *Physics of the Earth and Planetary Interiors*, 227, 1–19. https://doi.org/10.1016/j.pepi.2013.11.015
- Faccenda, M., & Capitanio, F. A. (2012). Development of mantle seismic anisotropy during subduction-induced 3-D flow. *Geophysical Research Letters*, 39(11). https://doi.org/10.1029/2012gl051988
- Faccenda, M., & Capitanio, F. A. (2013). Seismic anisotropy around subduction zones: Insights from three-dimensional modeling of upper mantle deformation and SKS splitting calculations. *Geochemistry, Geophysics, Geosystems*, 14(1), 243–262. https://doi.org/10.1002/ ggge.20055
- Faccenda, M., & Dal Zilio, L. (2017). The role of solid-solid phase transitions in mantle convection. *Lithos*, 268, 198–224. https://doi. org/10.1016/j.lithos.2016.11.007
- Faccenda, M., Ferreira, A. M., Tisato, N., Lithgow-Bertelloni, C., Stixrude, L., & Pennacchioni, G. (2019). Extrinsic elastic anisotropy in a compositionally heterogeneous Earth's mantle. *Journal of Geophysical Research: Solid Earth*, 124(2), 1671–1687. https://doi.org/10.1029/2018jb016482
- Faccenda, M., Minelli, G., & Gerya, T. (2009). Coupled and decoupled regimes of continental collision: Numerical modeling. *Earth and Planetary Science Letters*, 278(3–4), 337–349. https://doi.org/10.1016/j.epsl.2008.12.021
- Faccenna, C., Becker, T. W., Auer, L., Billi, A., Boschi, L., Brun, J. P., et al. (2014). Mantle dynamics in the Mediterranean. Reviews of Geophysics, 52(3), 283–332. https://doi.org/10.1002/2013rg000444
- Faccenna, C., Becker, T. W., Jolivet, L., & Keskin, M. (2013). Mantle convection in the middle east: Reconciling afar upwelling, Arabia indentation and Aegean trench rollback. *Earth and Planetary Science Letters*, 375, 254–269. https://doi.org/10.1016/j.epsl.2013.05.043
- Faccenna, C., Becker, T. W., Lucente, F. P., Jolivet, L., & Rossetti, F. (2001). History of subduction and back arc extension in the central Mediterranean. *Geophysical Journal International*, 145(3), 809–820. https://doi.org/10.1046/j.0956-540x.2001.01435.x
- Faccenna, C., Funiciello, F., Civetta, L., D Antonio, M., Moroni, M., & Piromallo, C. (2007). Slab disruption, mantle circulation, and the opening of the Tyrrhenian Basins. Special Papers—Geological Society of America, 418, 153. https://doi.org/10.1130/2007.2418(08)
- Faccenna, C., Mattei, M., Funiciello, R., & Jolivet, L. (1997). Styles of back-arc extension in the central Mediterranean. *Terra Nova*, 9(3), 126–130. https://doi.org/10.1046/j.1365-3121.1997.d01-12.x
- Faccenna, C., Piromallo, C., Crespo-Blanc, A., Jolivet, L., & Rossetti, F. (2004). Lateral slab deformation and the origin of the western Mediterranean arcs. *Tectonics*, 23(1). https://doi.org/10.1029/2002tc001488
- Funiciello, F., & Faccenna, C. (2010). Unraveling the geodynamics of the central Mediterranean in a tank. *Journal of the Virtual Explorer*, 36(1). https://doi.org/10.3809/jvirtex.2010.00231
- Gattacceca, J., & Speranza, F. (2002). Paleomagnetism of Jurassic to Miocene sediments from the Apenninic carbonate platform (southern Apennines, Italy): Evidence for a 60 counterclockwise Miocene rotation. *Earth and Planetary Science Letters*, 201(1), 19–34. https://doi.org/10.1016/s0012-821x(02)00686-6
- Geiss, E. (1987). A new compilation of crustal thickness data for the Mediterranean area. Annales Geophysicae Series B. Terrestrial and Planetary Physics, 5, 623–630.
- Gerya, T. (2019). Introduction to numerical geodynamic modelling. Cambridge University Press.
- Gerya, T., Connolly, J. A., & Yuen, D. A. (2008). Why is terrestrial subduction one-sided? *Geology*, 36(1), 43–46. https://doi.org/10.1130/g24060a.1
- Giacomuzzi, G., Civalleri, M., De Gori, P., & Chiarabba, C. (2012). A 3d vs model of the upper mantle beneath Italy: Insight on the geodynamics of central Mediterranean. *Earth and Planetary Science Letters*, 335, 105–120. https://doi.org/10.1016/j.epsl.2012.05.004
- Gueguen, E., Doglioni, C., & Fernandez, M. (1998). On the post-25 Ma geodynamic evolution of the western Mediterranean. *Tectonophysics*, 298(1–3), 259–269. https://doi.org/10.1016/s0040-1951(98)00189-9
- Guillaume, B., Funiciello, F., Faccenna, C., Martinod, J., & Olivetti, V. (2010). Spreading pulses of the Tyrrhenian Sea during the narrowing of the Calabrian slab. *Geology*, *38*(9), 819–822. https://doi.org/10.1130/g31038.1
- Guillaume, B., Husson, L., Funiciello, F., & Faccenna, C. (2013). The dynamics of laterally variable subductions: Laboratory models applied to the Hellenides. Solid Earth, 4(2), 179–200. https://doi.org/10.5194/se-4-179-2013
- Gutscher, M.-A., Malod, J., Rehault, J.-P., Contrucci, I., Klingelhoefer, F., Mendes-Victor, L., & Spakman, W. (2002). Evidence for active subduction beneath Gibraltar. *Geology*, 30(12), 1071–1074. https://doi.org/10.1130/0091-7613(2002)030<1071:efasbg>2.0.co;2
- Hansen, L. N., Faccenda, M., & Warren, J. M. (2021). A review of mechanisms generating seismic anisotropy in the upper mantle. *Physics of the Earth and Planetary Interiors*, 313, 106662. https://doi.org/10.1016/j.pepi.2021.106662
- Holt, A. F., Royden, L., & Becker, T. (2017). The dynamics of double slab subduction. *Geophysical Journal International*, 209(1), 250–265. https://doi.org/10.1093/gji/ggw496
- Holt, A. F., Royden, L. H., Becker, T. W., & Faccenna, C. (2018). Slab interactions in 3-D subduction settings: The Philippine Sea plate region. Earth and Planetary Science Letters, 489, 72–83. https://doi.org/10.1016/j.epsl.2018.02.024
- Hu, J., Faccenda, M., & Liu, L. (2017). Subduction-controlled mantle flow and seismic anisotropy in south America. Earth and Planetary Science Letters, 470, 13–24. https://doi.org/10.1016/j.epsl.2017.04.027
- Hua, Y., Zhao, D., & Xu, Y. (2017). P wave anisotropic tomography of the Alps. Journal of Geophysical Research: Solid Earth, 122(6), 4509–4528. https://doi.org/10.1002/2016jb013831



- Jagoutz, O., Royden, L., Holt, A. F., & Becker, T. W. (2015). Anomalously fast convergence of India and Eurasia caused by double subduction. *Nature Geoscience*, 8(6), 475–478. https://doi.org/10.1038/ngeo2418
- Jolivet, L., Augier, R., Robin, C., Suc, J.-P., & Rouchy, J. M. (2006). Lithospheric-scale geodynamic context of the Messinian salinity crisis. Sedimentary Geology, 188, 9–33. https://doi.org/10.1016/j.sedgeo.2006.02.004
- Jolivet, L., Faccenna, C., & Piromallo, C. (2009). From mantle to crust: Stretching the Mediterranean. Earth and Planetary Science Letters, 285(1–2), 198–209. https://doi.org/10.1016/j.epsl.2009.06.017
- Kaminski, E., Ribe, N. M., & Browaeys, J. T. (2004). D-rex, a program for calculation of seismic anisotropy due to crystal lattice preferred orientation in the convective upper mantle. *Geophysical Journal International*, 158(2), 744–752. https://doi. org/10.1111/j.1365-246x.2004.02308.x
- Karato, S.-I., Jung, H., Katayama, I., & Skemer, P. (2008). Geodynamic significance of seismic anisotropy of the upper mantle: New insights from laboratory studies. Annual Review of Earth and Planetary Sciences, 36, 59–95. https://doi.org/10.1146/annurev. earth.36.031207.124120
- Karato, S.-I., & Wu, P. (1993). Rheology of the upper mantle: A synthesis. Science, 260(5109), 771–778. https://doi.org/10.1126/ science.260.5109.771
- Kästle, E. D., El-Sharkawy, A., Boschi, L., Meier, T., Rosenberg, C., Bellahsen, N., et al. (2018). Surface wave tomography of the Alps using ambient-noise and earthquake phase velocity measurements. *Journal of Geophysical Research: Solid Earth*, 123(2), 1770–1792. https:// doi.org/10.1002/2017jb014698
- Katayama, I., & Karato, S.-I. (2008). Low-temperature, high-stress deformation of olivine under water-saturated conditions. *Physics of the Earth and Planetary Interiors*, 168, 125–133. https://doi.org/10.1016/j.pepi.2008.05.019
- Kaus, B., Le Breton, E., Reuber, G., & Schuler, C. (2020). Using geodynamic modeling to test plate tectonic scenarios of the Mediterranean-alpine area. In EGU General Assembly Conference Abstracts. (p. 11134).
- Kennett, B. (2009). Seismic wave propagation in stratified media. ANU Press.
- Király, Á., Faccenna, C., & Funiciello, F. (2018). Subduction zones interaction around the Adria microplate and the origin of the Apenninic arc. *Tectonics*, 37(10), 3941–3953. https://doi.org/10.1029/2018tc005211
- Koulakov, I., Kaban, M., Tesauro, M., & Cloetingh, S. (2009). P-and S-velocity anomalies in the upper mantle beneath Europe from tomographic inversion of ISC data. *Geophysical Journal International*, 179(1), 345–366. https://doi.org/10.1111/j.1365-246x.2009.04279.x
- Lo Bue, R., & Faccenda, M. (2019). Three-dimensional modelling of the recent evolution of the Mediterranean convergent margin. In AGU Fall Meeting Abstracts (Vol. 2019).
- Long, M. D., Hager, B. H., De Hoop, M. V., & van Der Hilst, R. D. (2007). Two-dimensional modelling of subduction zone anisotropy with application to southwestern Japan. *Geophysical Journal International*, 170(2), 839–856. https://doi.org/10.1111/j.1365-246x.2007.03464.x Lucente, F. P., Chiarabba, C., Cimini, G. B., & Giardini, D. (1999). Tomographic constraints on the geodynamic evolution of the Italian
- region. Journal of Geophysical Research, 104(B9). https://doi.org/10.1029/1999jb900147
- Lucente, F. P., & Margheriti, L. (2008). Subduction rollback, slab breakoff, and induced strain in the uppermost mantle beneath Italy. *Geology*, 36(5), 375–378. https://doi.org/10.1130/g24529a.1
- Lucente, F. P., Margheriti, L., Piromallo, C., & Barruol, G. (2006). Seismic anisotropy reveals the long route of the slab through the western-central Mediterranean mantle. *Earth and Planetary Science Letters*, 241(3–4), 517–529. https://doi.org/10.1016/j.epsl.2005.10.041
- Lucente, F. P., & Speranza, F. (2001). Belt bending driven by lateral bending of subducting lithospheric slab: Geophysical evidences from the northern Apennines (Italy). *Tectonophysics*, 337(1–2), 53–64. https://doi.org/10.1016/s0040-1951(00)00286-9
- Luth, S., Willingshofer, E., Sokoutis, D., & Cloetingh, S. (2013). Does subduction polarity changes below the Alps? Inferences from analogue modelling. *Tectonophysics*, 582, 140–161. https://doi.org/10.1016/j.tecto.2012.09.028
- Magni, V., Faccenna, C., van Hunen, J., & Funiciello, F. (2014). How collision triggers backarc extension: Insight into Mediterranean style of extension from 3-D numerical models. *Geology*, 42(6), 511–514. https://doi.org/10.1130/g35446.1
- Maino, M., Decarlis, A., Felletti, F., & Seno, S. (2013). Tectono-sedimentary evolution of the Tertiary Piedmont Basin (NW Italy) within the Oligo-Miocene central Mediterranean geodynamics. *Tectonics*, 32(3), 593–619. https://doi.org/10.1002/tect.20047
- Malinverno, A., & Ryan, W. B. (1986). Extension in the Tyrrhenian Sea and shortening in the Apennines as result of arc migration driven by sinking of the lithosphere. *Tectonics*, 5(2), 227–245. https://doi.org/10.1029/tc005i002p00227
- Margheriti, L., Lucente, F., & Pondrelli, S. (2003). SKS splitting measurements in the Apenninic-Tyrrhenian domain (Italy) and their relation with lithospheric subduction and mantle convection. *Journal of Geophysical Research: Solid Earth*, *108*(B4). https://doi. org/10.1029/2002jb001793
- Mason, W., Moresi, L., Betts, P., & Miller, M. (2010). Three-dimensional numerical models of the influence of a buoyant oceanic plateau on subduction zones. *Tectonophysics*, 483(1–2), 71–79. https://doi.org/10.1016/j.tecto.2009.08.021
- Mauffret, A., Frizon de Lamotte, D., Lallemant, S., Gorini, C., & Maillard, A. (2004). E-W opening of the Algerian basin (western Mediterranean). *Terra Nova*, *16*(5), 257–264. https://doi.org/10.1111/j.1365-3121.2004.00559.x
- Menke, W., & Levin, V. (2003). The cross-convolution method for interpreting SKS splitting observations, with application to one and two-layer anisotropic earth models. *Geophysical Journal International*, 154(2), 379–392. https://doi.org/10.1046/j.1365-246x.2003.01937.x Miller, M., & Piana Agostinetti, N. (2012). Insights into the evolution of the Italian lithospheric structure from s receiver function analysis.
- Earth Science Planetary Letters, 345–348, 49–59. https://doi.org/10.1016/j.epsl.2012.06.028
- Mishin, Y. A., Gerya, T. V., Burg, J.-P., & Connolly, J. A. (2008). Dynamics of double subduction: Numerical modeling. *Physics of the Earth and Planetary Interiors*, 171(1-4), 280–295. https://doi.org/10.1016/j.pepi.2008.06.012
- Nakagawa, T., Tackley, P. J., Deschamps, F., & Connolly, J. A. (2009). Incorporating self-consistently calculated mineral physics into thermochemical mantle convection simulations in a 3-D spherical shell and its influence on seismic anomalies in Earth's mantle. *Geochemistry, Geophysics, Geosystems*, 10(3). https://doi.org/10.1029/2008gc002280
- Neri, G., Marotta, A., Orecchio, B., Presti, D., Totaro, C., Barzaghi, R., & Borghi, A. (2012). How lithospheric subduction changes along the Calabrian arc in southern Italy: Geophysical evidences. *International Journal of Earth Sciences*, 101(7), 1949–1969. https://doi. org/10.1007/s00531-012-0762-7
- Neri, G., Orecchio, B., Totaro, C., Falcone, G., & Presti, D. (2009). Subduction beneath southern Italy close the ending: Results from seismic tomography. Seismological Research Letters, 80(1), 63–70. https://doi.org/10.1785/gssrl.80.1.63
- Panza, G., Pontevivo, A., Chimera, G., Raykova, R., & Aoudia, A. (2003). The lithosphere-asthenosphere: Italy and surroundings. *Episodes*, 26(3), 169–174. https://doi.org/10.18814/epiiugs/2003/v26i3/003
- Park, J., & Levin, V. (2002). Seismic anisotropy: Tracing plate dynamics in the mantle. *Science*, 296(5567), 485–489. https://doi.org/10.1126/science.1067319



- Patacca, E., Sartori, R., & Scandone, P. (1990). Tyrrhenian basin and Apenninic arcs: Kinematic relations since late Tortonian times. *Memorie della Societa Geologica Italiana*, 45, 425–451.
- Peral, M., Ruh, J., Zlotnik, S., Funiciello, F., Fernàndez, M., Vergés, J., & Gerya, T. (2020). Analog and numerical experiments of double subduction systems with opposite polarity in adjacent segments. *Geochemistry, Geophysics, Geosystems*, 21(6), e2020GC009035. https:// doi.org/10.1029/2020gc009035
- Petrescu, L., Pondrelli, S., Salimbeni, S., Faccenda, M., Group, A. W., et al. (2020). Mantle flow below the central and greater Alpine region: Insights from SKS anisotropy analysis at AlpArray and permanent stations. *Solid Earth*, *11*(4), 1275–1290. https://doi.org/10.5194/ se-11-1275-2020
- Piromallo, C., & Morelli, A. (1997). Imaging the Mediterranean upper mantle by P-wave travel time tomography. *Annals of Geophysics*, 40(4). https://doi.org/10.4401/ag-3890
- Piromallo, C., & Morelli, A. (2003). P wave tomography of the mantle under the Alpine-Mediterranean area. Journal of Geophysical Research: Solid Earth, 108(B2). https://doi.org/10.1029/2002jb001757
- Plomerová, J., Margheriti, L., Park, J., Babuška, V., Pondrelli, S., Vecsey, L., & Salimbeni, S. (2006). Seismic anisotropy beneath the northern Apennines (Italy): Mantle flow or lithosphere fabric? *Earth and Planetary Science Letters*, 247(1–2), 157–170. https://doi.org/10.1016/j. epsl.2006.04.023
- Presti, D., Totaro, C., Neri, G., & Orecchio, B. (2019). New earthquake data in the Calabrian subduction zone, Italy, suggest revision of the presumed dynamics in the upper part of the subducting slab. *Seismological Research Letters*, *90*(5), 1994–2004. https://doi. org/10.1785/0220190024
- Ranalli, G. (1995). Rheology of the Earth. Springer Science & Business Media.
- Rappisi, F., & Faccenda, M. (2019). Geodynamic and seismological numerical modelling for seismic anisotropy studies. In American Geophysical Union Fall Meeting.
- Richter, F. M. (1973). Finite amplitude convection through a phase boundary. Geophysical Journal International, 35(1-3), 265-276.
- Rollet, N., Déverchere, J., Beslier, M.-O., Guennoc, P., Réhault, J.-P., Sosson, M., & Truffert, C. (2002). Back arc extension, tectonic inheritance, and volcanism in the Ligurian Sea, western Mediterranean. *Tectonics*, 21(3), 6-1–6-23. https://doi.org/10.1029/2001tc900027
- Rosenbaum, G., Gasparon, M., Lucente, F. P., Peccerillo, A., & Miller, M. S. (2008). Kinematics of slab tear faults during subduction segmentation and implications for Italian magmatism. *Tectonics*, 27(2). https://doi.org/10.1029/2007tc002143
- Rosenbaum, G., Lister, G. S., & Duboz, C. (2002a). Relative motions of Africa, Iberia and Europe during alpine orogeny. *Tectonophysics*, 359(1–2), 117–129. https://doi.org/10.1016/s0040-1951(02)00442-0
- Rosenbaum, G., Lister, G. S., & Duboz, C., (2002b). Reconstruction of the tectonic evolution of the western Mediterranean since the Oligocene. Journal of the Virtual Explorer, 8. https://doi.org/10.3809/jvirtex.2002.00053
- Salimbeni, S., Pondrelli, S., & Margheriti, L. (2013). Hints on the deformation penetration induced by subductions and collision processes: Seismic anisotropy beneath the Adria region (central Mediterranean). *Journal of Geophysical Research: Solid Earth*, *118*(11), 5814–5826. https://doi.org/10.1002/2013jb010253
- Scarfi, L., Barberi, G., Barreca, G., Cannavò, F., Koulakov, I., & Patanè, D. (2018). Slab narrowing in the central Mediterranean: The Calabro-Ionian subduction zone as imaged by high resolution seismic tomography. *Scientific Reports*, 8(1), 1–12. https://doi.org/10.1038/ s41598-018-23543-8
- Schmid, C., Van Der Lee, S., & Giardini, D. (2004). Delay times and shear wave splitting in the Mediterranean region. *Geophysical Journal International*, 159(1), 275–290. https://doi.org/10.1111/j.1365-246x.2004.02381.x
- Schubert, G., Yuen, D. A., & Turcotte, D. L. (1975). Role of phase transitions in a dynamic mantle. *Geophysical Journal International*, 42(2), 705–735.
- Selvaggi, G., & Chiarabba, C. (1995). Seismicity and P-wave velocity image of the southern Tyrrhenian subduction zone. Geophysical Journal International, 121(3), 818–826. https://doi.org/10.1111/j.1365-246x.1995.tb06441.x
- Sieminski, A., Paulssen, H., Trampert, J., & Tromp, J. (2008). Finite-frequency SKS splitting: Measurement and sensitivity kernels. Bulletin of the Seismological Society of America, 98(4), 1797–1810. https://doi.org/10.1785/0120070297
- Spakman, W. (1990). Tomographic images of the upper mantle below central Europe and the Mediterranean. *Terra Nova*, 2(6), 542–553. https://doi.org/10.1111/j.1365-3121.1990.tb00119.x
- Spakman, W. (1991). Delay-time tomography of the upper mantle below Europe, the Mediterranean, and Asia minor. *Geophysical Journal International*, 107(2), 309–332.
- Spakman, W., van der Lee, S., & van der Hilst, R. (1993). Travel-time tomography of the European-Mediterranean mantle down to 1400 km. *Physics of the Earth and Planetary Interiors*, 79(1–2), 3–74. https://doi.org/10.1016/0031-9201(93)90142-v
- Spakman, W., & Wortel, R. (2004). A tomographic view on western Mediterranean geodynamics. In W. Cavazza, & G. M. Stampfli (Eds.), The TRANSMED Atlas. The Mediterranean region from crust to mantle. (pp. 31–52). Springer. https://doi.org/10.1007/978-3-642-18919-7_2
- Tackley, P. J. (1993). Effects of strongly temperature-dependent viscosity on time-dependent, three-dimensional models of mantle convection. Geophysical Research Letters, 20(20), 2187–2190. https://doi.org/10.1029/93gl02317
- Turcotte, D., & Schubert, G. (2014). Geodynamics. (Vol. 160-228, pp. 263-334). Cambridge University Press.
- VanderBeek, B., & Faccenda, M. (2021). Imaging upper mantle anisotropy with teleseismic P-wave delays: Insights from tomographic reconstructions of subduction simulations. *Geophysical Journal International*, 225(3), 2097–2119. https://doi.org/10.1093/gji/ggab081
- Van der Meer, D. G., Van Hinsbergen, D. J., & Spakman, W. (2018). Atlas of the underworld: Slab remnants in the mantle, their sinking history, and a new outlook on lower mantle viscosity. *Tectonophysics*, 723, 309–448. https://doi.org/10.1016/j.tecto.2017.10.004
- Van Hinsbergen, D. J., Torsvik, T. H., Schmid, S. M., Maţenco, L. C., Maffione, M., Vissers, R. L., et al. (2020). Orogenic architecture of the Mediterranean region and kinematic reconstruction of its tectonic evolution since the Triassic. Gondwana Research, 81, 79–229. https:// doi.org/10.1016/j.gr.2019.07.009
- van Hinsbergen, D. J., Vissers, R. L., & Spakman, W. (2014). Origin and consequences of western Mediterranean subduction, rollback, and slab segmentation. *Tectonics*, *33*(4), 393–419. https://doi.org/10.1002/2013tc003349
- Wortel, M., & Spakman, W. (2000). Subduction and slab detachment in the Mediterranean-Carpathian region. *Science*, 290(5498), 1910–1917. https://doi.org/10.1126/science.290.5498.1910
- Wortel, R., Govers, R., & Spakman, W. (2009). Continental collision and the step-wise evolution of convergent plate boundaries: From structure to dynamics. In S. Lallemand, & F. Funiciello (Eds.), *Subduction zone geodynamics* (pp. 47–59). Springer. https://doi.org/10.1007/978-3-540-87974-9_3
- Zhao, L., Paul, A., Malusà, M. G., Xu, X., Zheng, T., Solarino, S., et al. (2016). Continuity of the Alpine slab unraveled by high-resolution p wave tomography. *Journal of Geophysical Research: Solid Earth*, 121(12), 8720–8737. https://doi.org/10.1002/2016jb013310



- Zhong, S., & Gurnis, M. (1994). Role of plates and temperature-dependent viscosity in phase change dynamics. *Journal of Geophysical Research: Solid Earth*, 99(B8), 15903–15917. https://doi.org/10.1029/94jb00545
- Zhou, Q., Hu, J., Liu, L., Chaparro, T., Stegman, D. R., & Faccenda, M. (2018). Western US seismic anisotropy revealing complex mantle dynamics. Earth and Planetary Science Letters, 500, 156–167. https://doi.org/10.1016/j.epsl.2018.08.015
- Zhu, H., & Tromp, J. (2013). Mapping tectonic deformation in the crust and upper mantle beneath Europe and the north Atlantic ocean. Science, 341(6148), 871–875. https://doi.org/10.1126/science.1241335